



University of Dundee

Correlates of Patients' Preference for Therapeutic Injections in a Major Regional Referral Hospital in Ghana

Alhassan, Robert Kaba; Adu-Gyamfi, Bismark Appiah; Oliver, Agbolosu; Ayensu, Bright; Awoenam, Gbekor; Angela, Owusu

DOI:

10.1177/2158244018804583

Publication date: 2018

Licence: CC BY

Document Version
Publisher's PDF, also known as Version of record

Link to publication in Discovery Research Portal

Citation for published version (APA):

Alhassan, R. K., Adu-Gyamfi, B. A., Oliver, A., Ayensu, B., Awoenam, G., Angela, O., Nketiah-Amponsah, E., & Mwini-Nyaledzigbor, P. P. (2018). Correlates of Patients' Preference for Therapeutic Injections in a Major Regional Referral Hospital in Ghana: Implications for Policy and Clinical Practice. *SAGE Open*, 8(4), 1-8. https://doi.org/10.1177/2158244018804583

General rights

Copyright and moral rights for the publications made accessible in Discovery Research Portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

lake down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Download date: 11. May. 2024



Correlates of Patients' Preference for Therapeutic Injections in a Major Regional Referral Hospital in Ghana: Implications for Policy and Clinical Practice

SAGE Open October-December 2018: 1–8 © The Author(s) 2018 DOI: 10.1177/2158244018804583 journals.sagepub.com/home/sgo



Robert Kaba Alhassan , Bismark Appiah Adu-Gyamfi, Agbolosu Oliver, Bright Ayensu, Gbekor Awoenam, Owusu Angela, Edward Nketiah-Amponsah, and Prudence P. Mwini-Nyaledzigbor

Abstract

Abuse of injections, particularly in resource poor countries, remains a challenge evident in the increasing preference for therapeutic injections over oral medication. Objective of this study is to explore factors associated with patients' preference for therapeutic injection over oral medication in the Volta Regional Hospital, Ho in Ghana. The study is a cross-sectional survey conducted among 200 patients accessing care in Volta Regional Hospital. Data were analyzed using STATA statistical software for data analysis. Univariate probit regression was used to ascertain factors associated with patients' preference for therapeutic injections over oral medication (main outcome variable of interest). It was found that 74% of the 200 respondents preferred injection to oral medication. More outpatients preferred injectable medication over oral (p = .041); 86% of the respondents said they never experienced complication related to injectable medication. Patients who perceived injection as more efficacious were more likely to opt for it over oral medication (coefficient = 2.22; SE = 0.33; p < .05). It is concluded that patients' preference for injectable medication over oral remains high in Ghana, and this preference is significantly associated with patients' perceptions on superiority of injections over oral medication. There is the need to intensify health education for clients and in-service trainings for health providers to control abuse of therapeutic injections and promote patient safety.

Keywords

therapeutic injection, patients, preference, nursing, oral medication, Volta Regional Hospital, Ho, policy, clinical practice, Ghana

Plain Language Summary

Abuse of injections, particularly in poor countries, remains a challenge because of the increasing preference for therapeutic injections over oral medication. This research work sought to find out why health care clients prefer injections to oral medications in a major referral hospital in the Volta Region of Ghana.

At the end of the research, it was discovered that two thirds of patients visiting the Volta Regional Hospital (VRH) preferred therapeutic injection to oral medication. Predominant reasons for preference of injection were perceived quicker action, difficulty with oral medication compliance, and addiction to injectable medication. Also, patients who perceived therapeutic injection as more efficacious were more likely to opt for injections over oral medications.

Based on these results, the researchers recommend the need to increase public education for clients and health providers to control the indiscriminate use of injections in health care settings toward promoting the safety of patients.

Introduction

Increasing preference for therapeutic injection over oral medication remains a public health concern to health systems across the globe, especially in resource poor settings such as Ghana. Existing knowledge gaps on medical information among patients coupled with insufficient implementation of

School of Nursing and Midwifery, University of Health and Allied Sciences, Ho, Ghana

Corresponding Author:

Robert Kaba Alhassan, School of Nursing and Midwifery, University of Health and Allied Sciences, PMB 31, Ho, Volta Region, Ghana. Email: arkabason@gmail.com

²Department of Economics, University of Ghana, Legon, Ghana

health policies aggravates this challenge. According to a World Health Organization (WHO, 2015a) report, every year, billions of injections are used worldwide and over 95% are for therapeutic or pharmacological purposes. Even though therapeutic injection administration is a global phenomenon, its abuse particularly in developing countries remains a challenge evident in the increasing preference for therapeutic injections over oral medication (Poulos et al., 2016). Empirical studies have showed that about 70% of therapeutic injections administered are unnecessary as oral medications could have worked in most cases (Adejumo & Dada, 2013; Fox et al., 2013; WHO, 2015a, 2015b).

Notwithstanding increasing empirical evidence of the side effects and complications associated with therapeutic injections (Onyemocho, Anekoson, & Pius, 2013; WHO, 2015b), some health care clients prefer them to oral medications with varying reasons.

Patients' preference for therapeutic injections has been attributed to a number of factors including, perceived efficacy over oral formulations and economic benefits of therapeutic injections to providers.

Other factors associated with increasing preference for injectable medications are the weak regulatory policies on therapeutic injection practices in some countries, especially in Africa (Menzin et al., 2013). In Ghana for instance, even though the Ministry of Health (MoH) has internal protocols and guidelines on rational use of drugs including therapeutic injections (MoH, 2014), adherence to these protocols by clients and health care professionals is largely low.

A number of empirical studies have come out with varying conclusions on patients' preference for therapeutic injection¹ or oral medication² (Abdulebar, Tesfaye, Gedif, & Gebre-Mariam, 2007; Al Kanaani, Mahmud, Kouyoumjian, & Abu-Raddad, 2018; Nsimba, Gesase, & Massele, 2011; Nsimba, Massele, & Kayombo, 2011). However, many of these studies were conducted outside Ghana, hence the motivation for this current study. For instance, some previous studies suggest that patients measure the quality of health care they receive by the number of therapeutic injections prescribed and administered (Alhassan et al., 2015; Alhassan et al., 2016; Kermode, 2004; Kotwal et al., 2004).

Furthermore, there is dearth of empirical studies on patients' preference for treatment choice, particularly, regarding injectable and oral medications in Ghana. This study thus sought to explore determinants of patients' preference for therapeutic injections over oral medications in the VRH in Ho, Ghana. It is expected findings of this study would help inform policy decisions on client education and enforcement of standard professional practices on rational use of drugs in Ghana and beyond.

Method

Study Design, Population, and Setting

This is a descriptive cross-sectional survey conducted in the VRH, Ho in the Volta region of Ghana. As at 2014, the VRH

had a 240 bed capacity, comprising 42 units/departments and staff strength of 542. These staff comprised 232 nursing professionals, 23 medical doctors, and 287 paramedical staff. The VRH had bed turnover rate of 29.3%, and bed occupancy rate of 72.9% (VRH, 2014).

Sampling and Sample Size Determination

Purposive sampling method was used to select inpatients and outpatients who have had therapeutic injection administered to them at least in the last 6 months. The sample frame for this study was inpatients and outpatients accessing health care at the VRH. Based on Krejcie and Morgan (1970) formula³ for sample size determination for known population, a sample size of 205 was deemed representative of the estimated population of 440 inpatients and outpatients who attend the VRH on daily basis. Subsequently, 205 structured questionnaires were administered to the study participants who met the inclusion criteria.

Inclusion and Exclusion Criteria

All inpatients and outpatients attending the VRH and have had an injectable and oral medication experience in the last 6 months were included in the study. Patients who were also 18 years or more were included in the study. Exclusion criteria included patients who refused to consent voluntarily for the study. Likewise, minors below the ages of 18 years were excluded from the study. Finally, patients who were in critical chronic and emergency medical conditions were not considered because of their health status.

Data Collection

Questionnaire were piloted using 10 patients at the Ho Polyclinic to check for ambiguity, to ensure clarity of questions and relevance of responses. Data collection was done by trained research assistants from the School of Nursing and Midwifery, University of Health and Allied Sciences, Ho, for nearly a month.

Out of the 205 administered questionnaires, 200 were successfully retrieved with complete information representing approximately 98% return rate. The data exported to STATA were further disaggregated into two subsamples of inpatients (n = 50, 25%) and outpatients (n = 150, 75%).

Data Analysis

Data from the field was coded, captured, and cleaned using Microsoft Excel Spreadsheet and subsequently exported to STATA statistical software (version 12.0) for the data analysis. Data coding was done to ensure anonymity of respondents. To promote internal validity of the data collection tool, all questions were informed by research objectives and reviewed literature.

Alhassan et al.

Table 1. Summary Statistics on Patients' Sociodemographic Characteristics.

			M (SD) M (SD) M (SD) M (SD) 1) 33.4 (13.5) 33.2 (12.9) f (%) f (%) 0) 60 (30.0) 82 (41.0) 0) 90 (45.0) 118 (59.0) 5) 130 (65) 177 (88.5) 20 (10.0) 23 (11.5) 0) 50 (25.0) 74 (37.0) 0) 100 (50.0) 126 (63.0) 0) 142 (71.0) 188 (94.0)			
		Inpatients $(n = 50)$		Total	p value	
Characteristics	Observations	M (SD)	M (SD)	M (SD)		
Age	200	32.8 (11.1)	33.4 (13.5)	33.2 (12.9)	.8,132	
Gender	Observations	f (%)	f (%)	f (%)		
Males	200	22 (11.0)	, ,	` '	.368	
Females		28 (14.0)	90 (45.0)	118 (59.0)		
Religion						
Christian	200	47 (23.5)	130 (65)	177 (88.5)	.472	
Other		3 (1.5)	20 (10.0)	23 (11.5)		
Geographic location						
Rural	200	24 (12.0)	50 (25.0)	74 (37.0)	.046 [†]	
Urban		26 (13.0)	100 (50.0)	126 (63.0)		
Education						
Formal	200	46 (23.0)	142 (71.0)	188 (94.0)	.349	
Informal		4 (2.0)	8 (4.0)	12 (6.0)		
Marital status		,	,	. ,		
Married	200	27 (13.5)	65 (32.5)	92 (46.0)	.050	
Not married		23 (11.5)	85 (42.5)	108 (54.0)		
Employment status		` ,	,	,		
Employed	200	36 (18.0)	78 (39.0)	114 (57.0)	.010 [†]	
Unemployed		14 (7.0)	72 (36.0)	86 (43.0)		

Source. Field Data (2017).

Note. n = number of valid responses; f = frequency.

Parametric (univariate probit regression, VCE robust, and independent *t*-test) and nonparametric (chi-square and Fisher's exact) tests were used in the data analysis to test various hypotheses as appropriate. Multicollinearity diagnosis was conducted on all the explanatory variables prior to their inclusion in the regression model. The average variance inflation factor (VIF) was found to be 1.31, which is below the 10.0 rule of thumb necessary for exclusion from the regression model (Alhassan et al., 2016; Greene, 2002; Tabachnick & Fidell, 2001).

Findings

Results from the field data showed no significant difference in the average age of inpatients (32.8; SD=11.1) and outpatients (33.4; SD=13.5); little more than 60% of the patients interviewed were females, and the remaining were males; 63% of outpatients were from urban areas compared with 37% who were from rural areas (p < .05). Table 1 shows other sociodemographic details of respondents.

On exposure to the rapeutic injection, it was found that a significant proportion of outpatients (44.6%) had no exposure to the rapeutic injection compared with 18% of inpatients (p = .003); moreover, 24% of inpatients indicated they received

therapeutic injection more than three times in the last 1 month compared with 8.8% of outpatients (p < .05; see Table 2).

Overall, 49 (24.5%) of the respondents indicated they received therapeutic injectable medication outside a health facility environment. Out of this number, majority were inpatients who mostly patronized pharmacies (80%), chemical shops (10%), and home-based personal arrangements (1%; p < .05; see Table 2).

Out of the 200 valid responses, it was found that 74% said they preferred therapeutic injection to oral medication. More than 50% of the outpatients said they preferred therapeutic injection to oral medication compared with 19.5% of inpatients (p = .041). A further univariate probit regression test however showed these differences are not statistically significant even though the Fisher's exact test suggests otherwise. The predominant reasons given by respondents who preferred injectable medications were perceived quicker action (50%), difficulty with oral medication compliance (31%), and addiction to injection (3%; see Table 3). Majority (86%) of the respondents indicated they have never experienced a complication related to the rapeutic injection; 14% said they have suffered complications from injectable medication. Table 3 shows details of patients' experiences with complications related to therapeutic injections.

 $^{^{\}dagger}$ I-sided Fisher's exact (p < .05).

Table 2. Experiences of Patients With Therapeutic Injections.

		patients n = 50)	Outpatients $(n = 150)$		
Experiences	Observations	Proportion (95% CI)	Observations	Proportion (95% CI)	p value
Exposure to therapeutic injection ^a					
None	198	18.0 [7.1, 28.8]	198	44.6 [36.5, 52.7]	.003 [†]
Once	198	24.0 [12.0, 36.0]	198	21.6 [14.9, 28.3]	
2 times	198	22.0 [10.3, 33.7]	198	18.2 [12.0, 24.5]	
3 times	198	12.0 [2.8, 21.2]	198	6.8 [2.7, 10.8]	
More than 3 times	198	24.0 [12.0, 36.0]	198	8.8 [4.2, 13.4]	
Knowledge on reason for administered injection	48	75.0 [56.8, 93.2]	48	75.0 [56.8, 93.2]	.630
Received injection outside health facility	197	18.4 [7.3, 29.4]	197	27.0 [19.8, 34.3]	.152
Sources of therapeutic injection outside health	facility	-			
Pharmacy	49	80.0[53.2, 106.8]	49	41.0 [25.0, 57.1]	.078
Chemical shop	49	10.0[-10.1, 30.1]	49	5.0 [-2.1, 12.3]	
Home	49	1.0[-10.1, 30.1]	49	48.7 [32.4, 65.0]	

Source. Field Data (2017).

Note. n = number of valid responses; CI = confidence interval.

Table 3. Preferences for Therapeutic Injections Over Oral Medications.

		Patient category			
	Observations	Inpatients $(n = 50)$	Outpatients $(n = 150)$	Total	p value
Perspectives		f (%)	f (%)	f (%)	
Medication preference	200				
Injectable medication		38 (19.0)	110 (55.0)	148 (74.0)	.041 [†]
Oral medication		12 (6.0)	39 (19.5)	51 (25.5)	
Missing system		_	_	I (0.5)	
Reasons for preferring injection	148				
Quicker action		21 (14.0)	54 (36.0)	75 (50.0)	.058
Cheaper		0 (0.0)	0 (0.0)	0 (0.0)	
Addiction		0 (0.0)	4 (3.0)	4 (3.0)	
Defaulting in oral medications		18 (12.0)	28 (19.0)	46 (31.0)	
Missing system		-	-	23 (16.0)	
Complications due to injection	200				
Yes		4 (2.0)	24 (12.0)	28 (14.0)	.117
No		46 (23.0)	126 (63.0)	172 (86.0)	
Type of injection complication	28				
Abscess		9 (32.0)	8 (29.0)	17 (61.0)	.127
Allergy		4 (14.0)	5 (18.0)	9 (32.0)	
Paralysis		l (3.5)	0 (0.0)	l (3.5)	
Other		0 (0.0)	I (3.5)	I (3.5)	
Source of injection complication	28	. ,		. ,	
Government facility		10 (35.7)	10 (35.7)	20 (71.4)	.484
Private facility		4 (14.3)	4 (14.3)	8 (28.6)	

Source. Field Data (2017).

Note. n = number of valid responses; f = frequency.

^aNumber of times patients received therapeutic injection in the last 6 months.

 $^{^{\}dagger}$ I-sided Fisher's exact (p < .05).

[†]I-sided Fisher's exact (p < .05).

Alhassan et al. 5

Table 4. Univariate Probit Regression: Factors Associated With Preference for Therapeutic Injection.

	Dependent variable: Medication preference (therapeutic injection)			
Independent variables	Coefficient	SE (robust)	CI	
Type of patient				
Inpatient	0.53	0.23	[0.09, 0.98]	
Outpatient	1.0	1.0	[1.0, 1.0]	
Perceived therapeutic injection efficacy				
Yes	2.22^{\dagger}	0.33	[1.58, 2.87]	
No	1.0	1.0	[1.0, 1.0]	
Experienced therapeutic injection complication				
Yes	-0.20	0.31	[-0.81, 0.41]	
No	1.0	1.0	[1.0, 1.0]	
Gender				
Male	0.03	0.22	[-0.38, 0.47]	
Female	1.0	1.0	[1.0, 1.0]	
Geographic location				
Rural	-0.12	0.22	[-0.55, 0.32]	
Urban	1.0	1.0	[1.0, 1.0]	
Religion				
Christian	0.67	0.36	[-0.05, 1.38]	
Other	1.0	1.0	[1.0, 1.0]	
Education			_	
Educated	0.64	0.43	[-0.20, 1.48]	
Uneducated	1.0	1.0	[1.0, 1.0]	
Marital status				
Married	0.06	0.22	[-0.37, 0.48]	
Not married	1.0	1.0	[1.0, 1.0]	

Source. Field Data (2017).

Note. Probit regression: Wald $\chi^2(8) = 57.09$; prob $> \chi^2 = 0.0000$; pseudo $R^2 = .3368$; log pseudolikelihood = -90.812175; postestimation marginal effect (margin = 0.57; SE = 0.027; p = .000). CI = confidence interval.

It was also observed that patients who perceived therapeutic injection to be more efficacious than oral medication are more likely to opt for injectable medication (coefficient = 2.22; SE = 0.33; p < .05). Religious affiliation of patients also appeared to have an association with patients' preference for therapeutic injection over oral medication. It was observed that Christians are more likely to choose therapeutic injection than other forms of religious affiliations even though this association was not statistically significant (see Table 4). Likewise, factors such as gender, rural—urban location, education, and marital status were not statistically significant predictors of patients' preferences and choice of injectable medication over oral medication.

Discussion

It was found that contrary to study findings on United Kingdom (Mason, Crosson, Mason, & McGwin, 2017), 74% of the 200 respondents in this study said they preferred therapeutic injection to oral medication. Moreover, significant difference was found in the responses of inpatients and outpatients with more outpatients stating their preference

for therapeutic injection than their inpatient counterparts. Likewise, Fallowfield et al. (2006) in a study on United Kingdom found that nearly two thirds of the patients preferred oral tablet medication to therapeutic injection. Convenience and dislike of needles were reasons cited for patients' preference for oral tables, whereas the few who preferred therapeutic injection cited better adherence and personal preference.

In this study, it was observed that difficulty with oral medication compliance was mentioned as one of the key reasons for patients who preferred therapeutic injection. As found in similar studies (Fallowfield et al., 2006; Mason et al., 2017), perceived quicker action was mentioned by half of the patients who preferred therapeutic injection. These findings, appear to confirm the existing anecdotal claims that many patients in developing countries such as Ghana prefer therapeutic injection to oral medications, contrary to patients' health seeking behaviour in many developed countries (Fallowfield et al, 2006).

Albeit this study could not access relevant comparative literature, this divergence in findings might be attributed to the differences in study settings as developed countries with

 $^{^{\}dagger}p < .05$ (probit regression test, VCE (robust).

better health care systems are more likely to have better informed patients which could influence their health seeking behaviour and treatment choices.

However, it appears this assumption might not necessarily apply to all Western countries. Some studies in the Western world have revealed high preference for injectable medications by patients as compared with other treatment options. For instance, Mason et al. (2017) found greater proportions of diabetic patients who preferred therapeutic injection to laser treatment in the United States.

In terms of actual exposure to therapeutic injection, nearly all the 200 respondents (particularly inpatients) said they received therapeutic injection in the last 6 months. This observation could be attributed to the fact that many inpatients are likely to suffer from chronic and perhaps complicated conditions that demand more radical and invasive treatment options such as injections (Mason et al., 2017). Conversely, many outpatient conditions are usually less complicated and most probably managed with oral medications. In Ghana, this practice is in line with the MoH and Ghana Health Service (GHS) policy guidelines on rational use of drugs where invasive procedures such as therapeutic injections are discouraged for outpatient conditions (MoH, 2014).

It was also found in this study that only a few of the patients who ever received injectable medication knew of the reason for an injection administration. Moreover, significant number of the respondents (largely outpatients) indicated they ever received an injectable medication outside a health care facility. These revelations largely corroborate conclusions in previous studies where patients exhibited significant knowledge deficit on their medical treatment protocol partly due to knowledge asymetry between patients and medical professionals (Alhassan et al., 2015; Alhassan et al., 2016).

Also, reliance on unsafe outlets for therapeutic injection has been found to be prevalent, especially in Africa where poverty and limited medical knowledge compels health care clients to rely on these unsafe sources for therapeutic injection (Nsimba, Gesase, et al., 2011; Nsimba, Massele, et al., 2011). In this study, it was found that patients who took therapeutic injections outside a health care facility did so at either a pharmacy, chemical shop, or at home.

However, this study could not determine specific conditions that warranted the respondents to take their therapeutic injections at home as the case might be with chronic diabetic patients who are put on routine insulin therapeutic injection.

Experience of therapeutic injection–related complications was found to be minimal among the sampled patients; this observation could be attributed to the fact that the study site is a regional referral facility where there are more qualified health personnel which has the potential of reducing the risk of therapeutic injection complications (Alhassan et al., 2016; Obionu, 2007; Reid, 2010). Moreover, there are few published empirical literature on complications due to therapeutic injection in Ghana to form the basis for national level comparison with the current study.

On the determinants of patients' preference for therapeutic injections, significant differences were observed between inpatient and outpatients with the latter more likely to opt for therapeutic injection. This observation contradicts similar previous findings (Kotwal et al., 2004; Mason et al., 2017) where patients receiving inpatient care expressed higher preference for therapeutic injections than those who were not.

Also, this study found that patients who perceived high efficacy of therapeutic injection were more likely to choose therapeutic injection over oral medication similar to findings in previous studies where perceived efficacy was found to be a key determinant of patients' preference for therapeutic injection over other treatment options, especially in Africa (Kermode, 2004; Kotwal et al., 2004; Nsimba, Gesase, et al., 2011; Nsimba, Massele, et al., 2011).

Conclusion

Preference for injectable medication among patients remain high in Ghana as found in this study. It was observed that patients' perception of therapeutic injection efficacy significantly influence their chances of opting for therapeutic injection over oral medication. In light of this health-seeking behavior by patients, there is the need to intensify public education for patients and routine inservice training for health care providers to promote rational and safe use of therapeutic injections. The role of nursing professionals in this discourse is particulalry critical since therapeutic drug administration is a core nursing function in the clincal setting. Additionally existing policy regulations on rational use of drugs should be rigorously enforced to the latter to ensure high compliance by prescribers at the health service delivery points to promote patient safety and quality healthcare delivery.

Limitations

This study was conducted in a regional referral hospital which is likely to influence the caliber of patients visiting the facility and their experiences with therapeutic injections. Also, since hospital was used in the study it poses generalizability challenges to other regions in Ghana whose patients may have different health-seeking behavior because of sociocultural differences.

Finally, the use of purposive sampling technique to recruit the participants might have introduced some level of bias. Nonetheless, adopting a representative sample size based on the known study population gives credence to the findings of this study.

Suggestions for Future Research

Future studies could consider isolating the specific disease conditions of patients and their preference for therapeutic injection as this study did not delve into this nuance. Furthermore, future researchers could attempt a comparative Alhassan et al. 7

study of patients accessing care in private and public health care facilities to unearth potential differences in the health-seeking behavior of these patients. Finally, a nation-wide study involving selected health facilities in all the 10 regions of Ghana could help understand the regional and cultural dynamics in patients' preferences for treatment options.

Ethics Approval and Consent to Participate

Administrative ethical approval was given by the University of Health and Allied Science (UHAS) Research Ethics Committee (REC). Questionnaires administration were preceded by informed consents sought from each respondent. Respondents who were illiterates, thumb-printed on the consent form after reading and interpreting the content of the informed consent. Literate respondents however signed to consent for the study. The participants were informed of their right to participate or withdraw from the study without any consequences. Full disclosure of information to participants was made and their confidentiality and privacy maintained throughout the study by anomymising the data with unique codes.

Availability of Data and Materials

All data supporting our findings are contained in the manuscript. There are no restrictions to data sources per se but details of the full data may be accessed through one of the lead researchers, Dr. Robert Kaba Alhassan, email: arkabason@gmail.com or Mobile: +233, 226409.

Consent to Publish

All authors of this manuscript consented to publishing findings of this study

Acknowledgments

We acknowledge the support and cooperation of the inpatients and outpatients of the Volta Regional Hospital. We also express appreciation to the workers and management of the Volta Regional Hospital for granting us the permission to conduct the study in the hospital. We also express our appreciation to the management and faculty of the School of Nursing and Midwifery of the University of Health and Allied Sciences (UHAS) for their peer reviews and recommendations throughout this study. Last but not the least, the following 2016/2017 academic year Level 400 students of the Department of Nursing are acknowledged for their tireless efforts in the data collection, entry, and management. The students are Appiah Bismark Adu-Gyamfi, Gbekor Awoenam, Ayensu Bright, Owusu Angela, and Agbolosu Oliver.

Authors' Contributions

A.B.A., A.B., A.O., G.A., O.A., and R.K.A. initiated the conceptualization and writing of the paper. A.B.A., A.B., A.O., G.A., O.A., and R.K.A. collected, entered, and cleaned field data. R.K.A. and E.N. provided advice on the analysis and interpretation of results. R.K.A. and P.P.M. reviewed the manuscript and provided expert advice from the public health and health policy perspectives. All the authors read and approved the final manuscript.

Declaration of Conflicting Interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The author(s) received no financial support for the research, authorship, and/or publication of this article.

Notes

- Therapeutic injection in this context includes all form of injectable medications prescribed for therapeutic purposes. These injections include intramuscular, intradermal, subcutaneous routes, as per the World Health Organization (WHO, 2015a) guideline on the use of safety-engineered syringes.
- Oral medication in this study context includes all forms of medications administered by oral route such as tablets, capsules etc.
- 3. $s = \chi^2 NP (1 P) \div d^2 (N 1) + \chi^2 P (1 P)$ s = required sample size

 χ^2 = the table value of chi-square for 1 degree of freedom at the desired confidence level (3.841)

N = the population size

P = the population proportion (assumed to be .50 as this would provide the maximum sample size)

d = the degree of accuracy expressed as a proportion (.05)

ORCID iD

Robert Kaba Alhassan https://orcid.org/0000-0003-4227-4854

References

Abdulebar, S., Tesfaye, F., Gedif, T., & Gebre-Mariam, T. (2007). The prevalence and community perception of therapeutic injection use in Meskan district, South Central Ethiopia. *Ethiopian Pharmacology Journal*, 25, 63-71.

Adejumo, P. O., & Dada, F. A. (2013). A comparative study on knowledge, attitude, and practice of injection safety among nurses in two hospitals in Ibadan, Nigeria. *International Journal of Infection Control*, 9(1), 1-6.

Alhassan, R. K., Duku, S. O., Jassens, W., Nketiah-Amponsah, E., Spieker, N., van Ostenberg, P., . . . Rinke de Wit, T. F. (2015). Comparison of perceived and technical health care quality in primary health facilities: Implications for a sustainable national health insurance scheme in Ghana. *PLoS ONE*, 10(10), e0140109.

Alhassan, R. K., Nketiah-Amponsah, E., Spieker, N., Arhinful, D. K., Ogink, A., van Ostenberg, P., . . . Rinke de Wit, T. F. (2016). Effect of community engagement interventions on patient safety and risk reduction efforts in primary health facilities: Evidence from Ghana. *PLoS ONE*, 10(11), e0142389.

Al Kanaani, Z., Mahmud, S., Kouyoumjian, S. P., & Abu-Raddad, L. J. (2018). The epidemiology of hepatitis C virus in Pakistan: Systematic review and meta-analyses. *Royal Society Open Science*, 5, Article 180257.

Fallowfield, L., Atkins, L., Catt, S., Cox, A., Coxon, C., Langridge, C., & Price, M. (2006). Patients' preference for administration of endocrine treatments by injection or tablets: Results from a study of women with breast cancer. *Annals of Oncology*, 17, 205-210.

- Fox, R. J., Salter, A. R., Tyry, T., Sun, J., You, X., Laforet, G., & Campagnoglo, D. (2013). Treatment discontinuation and disease progression with injectable disease-modifying therapies: Findings from the North American research committee on multiple sclerosis database. *International Journal of MS Care*, 15, 194-201.
- Greene, W. H. (2002). Econometric analysis (5th ed.). New York, NY: Prentice Hall.
- Kermode, M. (2004). Unsafe therapeutic injections in low-income countries health setting: Need for therapeutic injection safety promotion to prevent the spread of blood-borne viruses. *Health Promotion International*, 19, 95-103.
- Kotwal, A., Priya, R., Thakur, R., Gupta, V., Kotwal, J., & Seth, T. (2004). Injection practices in metropolis of North India: Perceptions, determinants, and issues of safety. *Indian Journal of Medical Science*, 58, 334-344.
- Krejcie, K. V., & Morgan, D. W. (1970). Determining sample size for research activities. *Educational and Psychological Measurement*, 30, 607-610.
- Mason, L., Crosson, J. N., Mason, J. O., III, & McGwin, G., Jr. (2017). Patient preferences with regard to laser versus intravitreal injections in the treatment of diabetic macular edema. *Journal of Ophthalmology*, 2017, 1-4.
- Menzin, J., Caon, C., Nichols, C., White, L. A., Friedman, M., & Pill, M. W. (2013). Narrative review of the literature on adherence to disease-modifying therapies among patients with multiple sclerosis. *Journal of Managed Care Pharmacy*, 19(1, Suppl. A), S24-S40.
- Ministry of Health (MoH). (2014). *Ghana National Medicines Review Policy 2014*. Accra, Ghana. Retreived from: http://www.who.int/medicines/technical_briefing/tbs/Ghana_NDP 2014 REVIEW PROCESS.pdf.
- Nsimba, S. E. D., Gesase, A. P., & Massele, A. Y. (2011). Dangers of injections overuse in developing countries with a high HIV/ AIDS prevalence: A review on HIV risk hazards, traumatic effects and other blood borne infections. *Asian Pacific Journal* of *Tropical Disease*, 1, 158-163.
- Nsimba, S. E. D., Massele, A., & Kayombo, E. J. (2011). Reducing injection overuse through consumers'–prescribers' interactional group discussions in Dar es Salaam region, Tanzania. *Therapy*, 8(2), 179-187.
- Obionu, C. (2007). *Primary health care for developing countries* (2nd ed., p. 149). Enugu, Nigeria: Evans Publishers.
- Onyemocho, A., Anekoson, J. I., & Pius, E. O. (2013). Knowledge and practice of injection safety among workers of Nigerian prison service health facilities in Kaduna State. *American Journal of Public Health Research*, 1, 171-176.
- Poulos, C., Kinter, E., Yang, J., Bridges, J. F. P., Posner, J., & Reder, A. T. (2016). Patient preferences for injectable treatments for multiple sclerosis in the United States: A discretechoice experiment. *Patient*, 9, 171-180.
- Reid, S. (2010). Preliminary results of the World Health Organization (WHO) GBD from unsafe injections. Geneva, Switzerland: World Health Organization.

- Tabachnick, B. G., & Fidell, L. S. (2001). *Using multivariate analysis* (4th ed.). Boston, MA: Allyn & Bacon.
- Volta Regional Hospital. (2014). *Hospital administrative records*. Ho: Ghana Health Service, Volta Regional Hospital.
- World Health Organization. (2015a). WHO calls for worldwide use of "smart" syringes, Press release. Geneva. Retrieved from http:// www.who.int/mediacentre/news/releases/2015/injection-safety/en/
- World Health Organization. (2015b). WHO guideline on the use of safety-engineered syringes for intramuscular, intradermal and subcutaneous injections in health-care settings (WHO/HIS/SDS/2015.5). Retrieved from www.who.int/infection-prevention/campaigns/injections/pops-is meeting1.pdf.

Author Biographies

Robert Kaba Alhassan (RN, BA, MPhil, PhD) is a lecturer and a researcher at the University of Health and Allied Sciences, School of Nursing and Midwifery. Research interest include: patient safety, healthcare quality, healthcare policy, nursing education and practice.

Bismark Appiah Adu-Gyamfi (bachelor of nursing) is a graduate general nurse whose research interests include clinical nursing practice and safety use of medications.

Agbolosu Oliver (bachelor of nursing) completed the University of Health and Allied Sciences, School of Nursing and Midwifery in 2016/2017 academic year and now aspiring to be an academic in the area of nursing research and evidence-based nursing.

Bright Ayensu (bachelor of nursing) professional skills are in clinical nursing practice and client-centered quality of healthcare delivery in resource poor settings. Author's basic qualification Bachelor of Nursing.

Gbekor Awoenam (bachelor of nursing) is a general nurse by training and currently focused on bedside nursing hoping to build more research skills in nursing research and evidence-based practice.

Owusu Angela (bachelor of nursing) graduated with a bachelor of nursing degree from the University of Health and Allied Sciences, School of Nursing and Midwifery. Interest areas include general nursing practice, particularly medical and surgical nursing.

Edward Nketiah-Amponsah (BA, MPhil, PhD) is an associate professor of health economics at the Department of Economics, University of Ghana Legon. For several years his research interests include healthcare financing and cost efficiency analysis in healthcare delivery.

Prudence P. Mwini-Nyaledzigbor (RN, RM, PHN, BA, MPhil, PhD) is an associate professor of nursing and a former dean of the School of Nursing and Midwifery, University of Health and Allied Sciences, Ho. Research interests include, public health nursing, midwifery and neglected tropical diseases in Africa.